Author's response to reviews

Title: Planning, timing and quit success: Study protocol of a randomised controlled trial.

Authors:

Ron Borland (Ron.Borland@cancervic.org.au)
James Balmford (James.Balmford@cancervic.org.au)
Elena Swift (Elena.Swift@cancervic.org.au)

Version: 2 Date: 19 November 2012

Author's response to reviews:

20 November 2012

Dear editors,

Following a request we have made minor changes to the formatting of the recently-submitted manuscript "Planning, timing and quit success: Study protocol of a randomised controlled trial." for consideration for publication in BMC Public Health.

We have added the trial registration details to the end of the abstract, and also provided evidence of the funding source and ethical approval for the study as supplemental files. There are currently no other manuscripts under submission with other journals based on this study protocol.

All correspondence should be directed to:
Dr Ron Borland
VicHealth Centre for Tobacco Control
The Cancer Council Victoria
1 Rathdowne St Carlton VIC 3053
Melbourne, Australia
Tel: +61 3 9635 5185
Fax: +61 3 9635 5440
Email: Ron.Borland@cancervic.org.au

We thank you for considering the manuscript and look forward to a positive outcome.

Yours sincerely,

Ron Borland Ph.D